Notes for the RICH1 phone meeting 16/02/07

People present:

Bristol: Nick, Richard.
CERN: Kenneth, Chris, Erich

IC: Dave, Gabriel.

Oxford: Rohan.

RAL: Peter.

Apologies: Neville, Matthew, Derek, Richard P, Trevor, Bill.

Topics discussed:

1.  Gas Enclosure.

The diaphragm seal has been fitted to the exit window, after a measurement of the beam pipe alignment was made.  All this was achieved successfully.  The first round of leak testing has been conducted and no leaks from the diaphragm found.  

The RICH1 gas enclosure has been partially filled with  helium to 4mbar over pressure.  The 1/e decay time was 6 minutes, which indicates a leak of about 100 l/hour corresponding to a loss of all C4F10 within one year.  A leak rate of 1 l/hour would be deemed acceptable.

Last March/April the gas enclosure was tested for leaks with all its blanking plates on and the leak rate was less than 1 l/hour.  The VELO end of the seal to Velo has been leak tested and found to be tight.  The seal to Velo now replaces the front blanking plate and the exit window the back one.

Last week evidence was found for a leak on the seal to Velo and seal to RICH1 at 8 o’clock looking along the + z-direction.  However there is no quantitative measure of the size of the leak.

This week evidence for a leak at 9 o’clock on the seal to Velo, with a preference, but no overwhelming evidence, for it being at the Velo end was found.  The C-side Door was removed and helium squirted in from the outside in order to differentiate between leeks on the welds and on the o ring seals.  There is a preference for the o ring seals at the Velo end.

A helium filled balloon was fitted to the region with the leak, finding evidence for the leak at 9 o’clock and also one at one o’clock., which is difficult to access due to magnetic shielding.  As measurements have not been quantatitive there could still be further holes.  It should be possible to remove seal to Velo for further inspections, although this is a very delicate operation.

2. Quartz window.

The facilities used for the gluing of the RICH2 quartz windows will be available for the RICH 1 ones in April/May.  A similar procedure to that used for RICH2 will be used.

3. Spherical mirrors.

The four spherical mirrors and their frame have been received from CMA.  Measurements made by CMA indicate a good level of sphericity.  Measurements of the D0 and radius of curvature have been made at CERN with both quantities being well within specifications for all four mirrors.  

The mirrors were taken to SESO for coating on the 16/02/07 along with storage frames which will be needed for their protection and transportation once the reflective coating has been applied.  The coating should be finished in two to three weeks.  Fabio will obtain a more precise schedule when at SESO.

Manufacture of the optical alignment frame and installation tooling is on schedule and should be completed at the end of February and in CERN at the beginning of March.

Payment is to be made to CMA, who will also be asked to keep the mandrill for the mirror.

4.  Flat Mirrors

The same alignment frame referred to in item 3 will be ready at the end of February for the beginning of March.

5.  HPD Assembly

The frame for the upper box is being worked on and should be completed in a few weeks.  16 cooling plates have been completed.

1 column has completed and tested, 1 is ready for testing in the SSB (will be tested next week) and is 1 is having its HV soldered by Didier Friday 16/02/07 and is also to be tested next week
 

2 columns have had electronics mounted (HV, LV, L0 boards).  There should be enough HPD's at Cern to mount and test one of these by the end of next week.  More HPD's are due to arrive at the end of next week.
 

2 more have HV and LV and are awaiting L0 boards.  More L0 boards have been requested and should arrive some time next week also.
 

The overall schedule is to have 3-4 columns completed and tested by the end of next week, hopefully another 3 by the end of the week after so the first half of RICH-1 should be complete in around 2.5-3 weeks.
Mechanics of rail system:  Rohan has completed the modifications to accommodate the HPD box.  Should be in position to place order next week (week beginning 19/02/07).  A quotation from NAB of £13500 has been acquired with a 6 week delivery schedule. But 3 quotes are required for purchases over £5k.

7. Magnetic distortion monitoring system.

At the time of the last phone meeting with Syracuse, there was no definite proposal for a manufacturer.  They have been trying without success to outsource and have now obtained components and are planning to build the system, modifying the components if needed.  There is a backup plan, but nobody definitely assigned to undertaking it.  A further phone meeting will take place on 22/02/07.
